Author’s response to reviews

Title: The Roche Cancer Genome Database 2.0

Authors:

Jan Küntzer (jan.kuentzer@roche.com)
Daniela Eggle (daniela.eggle@roche.com)
Hans-Peter Lenhof (lenhof@bioinf.uni-sb.de)
Stefan Klostermann (stefan.klostermann@roche.com)
Helmut Burtscher (helmut.burtscher@roche.com)

Version: 3 Date: 12 May 2011

Author’s response to reviews:

As wished by the Editorial Production Team we changed the affiliations, added all email addresses, removed the highlighting, and finally checked for typos.

We changed the surname of one authors due to her marriage.

The structure is still the same as wished by the editorial team, and we hope we can stick with that.